Minutes of DRAGON Meeting July 26, 2005

Started 10:45 26/07/05

Present: JDA, TK, HC, BG, CV, DH, JP, BW, RL, AH, AO, JC, PM, GR, LF (Recorder).

Items:

1. Minutes from July 19, approved.

2. Business arising from minutes
PM is working on fixing the water leak on Q3

Also an “instant-off” system should be looked into for the DSSSD, using the leakage current signal (DH)

3. DRAGON Hardware

Pumps are off, triple alpha source has been returned to TUDA, solid target chamber is in.

O-16 run:
Targets need to be made and shipped to DRAGON for this run, which begins 4th August.

BGO need to be calibrated for 25MeV gammas.

Are they using the DSSSD? If so they will need help with this..


JDA: Possibly replace new DSSSD with a different one and use the cooling system, or possibly they could use the ion chamber instead. It is also possible they could buy their own detector.

Cooling may or may not be appropriate for the experiment; the group should be contacted for information regarding this matter.

JP: NIM module (LeCroy 49 fan-in fan-out) went down in the heavy ion trigger. It has been replaced.

JP: Looking at the problem with the tdc in the DSSSD. It appeared very noisy. Ran overnight to test and also tried swapping modules etc. Still not sure about the origins of the problem.


Possibly MCP sparks could cause this.


Possibly a software problem.

JC: Electronics grant. Looking into getting a VF 48 board to test, this is still under development though as it is cutting edge technology.

Other possibility is a FDGA programmable unit.

DH: Put in a request for the design of mount for iris aperture to go in the box upstream of the target.

Can probably buy the iris itself quite reasonably.

4. E989 Status

HC presented table of run info. It has been split into several smaller sheets, depending on data source/type:

Raw Run Sheet Data

Raw Histogram Data

Raw History Data

Calculated Ratios

Beam Normalization Factors.

DH: Perhaps the ratio of BGO trigger rate to Ge trigger rate could give a good indication of whether or not the fraction of the beam spilled varied a lot. Also could determine variation in the Na-26 contamination.

AO: Some plots would be useful to visualize the data.

LF presented table of Elastic Monitor info – a)FC1/FC4 ratios; some runs with quite low (compared to other runs) ratios.  Unsure what caused this, possibly bad transmission.

JC: Check histories and strip tool to confirm FC4 readings.

b)Best runs for calculating beam normalization constant, R. There are a group of runs here with low elastic counts/FC1 and FC4 ratios. Unsure about the cause of this too.

c)Error on FC1 and FC4 readings. This is found by assuming the error on elastic monitor counts is purely statistical and subtracting it from the error on the elastic monitor counts/FC ratios.

5. AOB

RL comments on TUDA run.

RL: They had proposed 10 different energy steps, got 4 done. Very grateful to DRAGON for use of the facility.

Note-FCCH bias goes off and also problem with slit positions and beam moving.

In last run saw a 0.45% difference between start and end energy measurements, this run had 6 foils and 4 buncher trips so perhaps this caused the energy change…

RL: Possibly more runs later.

DH: Yesterday Brian spoke of hydrogenous contaminants in the target, is this a serious problem?

RL: It does pose a problem; talk of new target production methods that don’t use water.

Ca-40 

CV: Would like to measure three strong resonances.


Possibly condition ED1 to 175kV…was at 172kV and DH was concerned about the excess anode current at this voltage.


This experiment needs the ion chamber; if it is needed for the O-16 experiment perhaps it could be put in now.


Also investigate possible pile-up rejection for the ion chamber.

